Author’s response to reviews

Title: Macronutrients, Vitamins and Minerals Intake and Risk of Esophageal Squamous Cell Carcinoma in a Population at Risk for Esophageal Cancer

Authors:

Mahsa Jessri (jessri@ualberta.ca)
Bahram Rashidkhani (b_rashidkhani@sbmu.ac.ir)
Bahareh Hajizadeh (bhhajizadeh@yahoo.com)
Maryam Jessri (m.jessri@uq.edu.au)
Carolyn Gotay (carolyn.gotay@ubc.ca)

Version: 2 Date: 19 July 2011

Author’s response to reviews: see over
Dear Professor Gabriel,

Greetings,

Thank you for providing us an opportunity to revise the manuscript entitled “Macronutrients, Vitamins and Minerals Intake and Risk of Esophageal Squamous Cell Carcinoma: a Case-Control Study in Iran” (Manuscript ID: 6714475951821561) for publication in the “Nutrition Journal”. We have taken each critique and comment very seriously, and have provided point-by-point response to the reviewer’s concerns. In addition the entire text of the manuscript has been reviewed, edited and improved. However, authors are willing to provide further modifications, in case it is needed.

Again we thank you for your consideration.

Sincerely,

A Prof. Bahram Rashidkhani, MD, MPH, PhD in Epidemiology

Community Nutrition Department, Faculty of Nutrition Sciences and Food Technology, National Nutrition and Food Technology Research Institute, Shahid Beheshti University of Medical Sciences and Health Services, Tehran, Iran

Phone: +98 21 22360658; Fax: +98-2122360657